\documentclass[11pt,a4paper]{article}
%\documentstyle[epsf,12pt]{article}
\usepackage{epsfig}
\usepackage{graphicx}
\usepackage{eurosym}
\usepackage{multicol}
\usepackage{subfig}
\usepackage{caption}
\usepackage{times}[11pt]
\usepackage{geometry}
\geometry{a4paper, portrait, margin=3cm}
%\textwidth = 15.59cm
%\hoffset = -0.54cm
%\topmargin = 0.5cm 
%\headheight = 0.5cm
%\headsep = 0.5cm
%\voffset = -0.54cm
%\oddsidemargin = 0.7cm
%\marginparsep = 0.2cm
%\marginparwidth = 1.0cm
\def\refname{------------------------}%<----------



\begin{document}
\thispagestyle{empty}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{center}
	{\Large \bf
	% Title of your contribution:
	Abstract template for contributions to FFK2023
	\bigskip\\}
	% Author list, presenting author :
	First Author$^{a,b}$, Authors$^{b}$, \underline{Underline Presenting Author}$^{a,c}$, Further Coauthors$^{c}$\\
	%Alternative to listing all coauthor is a 'on-behalf-of-collaboration-phrase':
	(alternative to coauthors) on behalf of the something collaboration\medskip\\
	% Affiliation(s), assignment needs to be done manually
	$^a$ {\em Affiliation1}\\
	$^b$ {\em Affiliation2}\\
	$^c$ {\em Affiliation3}\bigskip\\
\end{center}


% Your abstract text:
This is a \LaTeX\ template for submitting a contribution as a single-page pdf to FFK2023.
The use of this template is not mandatory and every abstract that fits on one page (DIN A4) and is attached as pdf during the abstract submission will be considered.
If the one-page space constraint can be fulfilled without excessive style modifications feel free to add figures (cf. Fig.~\ref{tmpfig}), tables (cf. Table~\ref{tmptab}), and/or references~\cite{tmpref} to your abstract and as many authors and affiliations as necessary.

\begin{multicols}{2}

{
\centering
\includegraphics[width=6cm]{FFK2023-Logo_reduSized.jpg}
\captionof{figure}{Logo of FFK2023} \label{tmpfig}
}
	
{
\centering
\begin{tabular}{c|cc}
Bank transfer     & Full & Student \\
(Cash on arrival) & fee  & fee \\
\hline
\hline
Oral or no        &  300 \euro  &  150 \euro  \\
presentation      & (350 \euro) & (200 \euro) \\
\hline 
Poster            &  250 \euro  &  100 \euro  \\
presentation      & (300 \euro) & (150 \euro) \\
\hline
\end{tabular}
\captionof{table}{Conference fee overview} \label{tmptab}
}

\end{multicols}	
	
%for single figures and/or tables	
%\begin{figure}[tbh]
%\epsfxsize=5cm
%%\centerline{\epsfbox{tmpfig.eps}}
%\center{\includegraphics[width=0.4\textwidth]{FFK2023-Logo_reduSized.jpg}}
%\caption{\label{tmpfig} Logo of FFK2023}
%\end{figure}

%\begin{table}[tbh]
%	\begin{center}
%		\begin{tabular}{c|cc}
%Bank transfer     & Full & Student \\
%(Cash on arrival) & fee  & fee \\
%\hline
%\hline
%Oral or no        &  300 \euro  &  150 \euro  \\
%presentation      & (350 \euro) & (200 \euro) \\
%\hline 
%Poster            &  250 \euro  &  100 \euro  \\
%presentation      & (300 \euro) & (150 \euro) \\
%\hline
%\end{tabular}
%	\end{center}
%	\caption{\label{tmptab} Conference fee overview}
%\end{table}

Please follow this format for the filename of your single pdf-page, where the name of the presenting author should be used: \textit{FFK2023-Abstract\_FirstnameLastname.pdf}.
Please upload your final pdf file during the abstract submission process on the indico page of FFK2023 and follow the instructions there~\cite{indicourl}.
Your abstract will be publicly available on indico.
In case you have any further questions please contact \textit{ffk2023-office@oeaw.ac.at} for some~\cite{pc}.

%References (<10)
\begin{thebibliography}{9.}
%\frenchspacing
\setlength{\itemsep}{0em}
\setlength{\parskip}{0em} 

\bibitem{tmpref}
T. M. Plate {\em et al}., Journal of Examples {\bf vol} (year) page(s).
\bibitem{indicourl}
https://indico.cern.ch/event/1164804/abstracts/
\bibitem{pc}
\textit{private communication}
 
\end{thebibliography}

\end{document}  